Author's response to reviews

Title: Mammoth Interatrial Septal Aneurysm in the ICE Age

Authors:

Ravinay Bhindi (ravinay.bhindi@cardiov.ox.ac.uk)
John Timperley (john.timperley@orh.nhs.uk)
Oliver J Ormerod (Oliver.ormerod@orh.nhs.uk)

Version: 4 Date: 18 August 2007

Author's response to reviews:

Dear Dr Picano

Thank you for your email dated 3 July 2007. We apologise for the stylistic inconsistencies in our earlier submission which have now been rectified.

1. A separate abstract is included after the title page in the required format.
2. The body of the manuscript is now formatted as per the journal requirements i.e: background, case presentation, conclusion and includes figures and references
3. Competing interests and authors' declarations have also been incorporated
4. The referencing style is now corrected

We hope that these changes are satisfactory and will allow the publication of this article.

Regards

Dr Bhindi